Author's response to reviews

Title: Measurement of the total antioxidant response using a novel automated method in subjects with nonalcoholic steatohepatitis

Authors:

Mehmet Horoz (mehmethoroz@yahoo.com)
Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Fusun F Bolukbas (fusunbol@yahoo.ca)
Tevfik Sabuncu (tsabuncu@harran.edu.tr)
Mehmet Aslan (m.aslan301@mynet.com)
Serpil Sarifakiogullari (serpilsarifaki@yahoo.com)
Necla Gunaydin (g.necla@mynet.com)
Ozcan Erel (erelozcan@harran.edu.tr)

Version: 2 Date: 11 March 2005

Author's response to reviews:

Dear Editor, March 11, 2005
After submission of our manuscript, we noticed that we didn't include the statistical analysis section to the main manuscript. Thus, we added statistical analysis section and we aimed to resubmit the main text of manuscript.
Sincerely Yours
Mehmet HOROZ, MD.